Author’s response to reviews

Title: Building a learning community of Australian clinical genomics: a social network study of the Australian Genomic Health Alliance

Authors:

Janet Long (janet.long@mq.edu.au)
Chiara Pomare (chiara.pomare@mq.edu.au)
Stephanie Best (stephanie.best@mq.edu.au)
Tiffany Boughtwood (tiffany.boughtwood@mcri.edu.au)
Kathryn North (kathryn.north@mcri.edu.au)
Louise Ellis (louise.ellis@mq.edu.au)
Kate Churruca (kate.churruca@mq.edu.au)
Jeffrey Braithwaite (jeffrey.braithwaite@mq.edu.au)

Version: 1 Date: 20 Jan 2019

Author’s response to reviews:

Dear Reviewers,

Thank you for your close reading and useful comments on the manuscript. We have attached a point by point response to each of the issues raised as Supplementary material "Response to reviewers BMED-D-18-01488_v2". Changes in the manuscript are highlighted in yellow.

Kind regards,

Janet Long